Author's response to reviews

Title: Long-term exposure to air pollution and hospital admissions for ischemic stroke. A register-based case-control study using modelled NOx as exposure proxy.

Authors:

Anna Oudin (anna.oudin@med.lu.se)
Emilie Stroh (emilie.stroh@med.lu.se)
Ulf Strömberg (ulf.stromberg@med.lu.se)
Kristina Jakobsson (kristina.jakobsson@med.lu.se)
Jonas Björk (jonas björk@skane.se)

Version: 11 Date: 14 August 2009

Author's response to reviews:

Dear Editors,

I discovered three typographical errors on page 7 and 8 that I corrected

Sincerely,

Anna Oudin